Author’s response to reviews

Title: The impact of self-reported exposure to whole-body-vibrations on the risk of disability pension among men: a 15 year prospective study

Authors:

Finn Tüchsen (ftu@nrcwe.dk)
Helene Feveile (hfe@nrcwe.dk)
Karl B. Christensen (K.Christensen@pubhealth.ku.dk)
Niklas Krause (nkrause@uclink4.berkeley.edu)

Version: 5 Date: 28 May 2010

Author's response to reviews:

To BMC Series Editorial Production Team

We have now made the requested changes as shown below

Best wishes
Finn Tüchsen

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is ?First name Initial Last name? (e.g. Joe F Bloggs).

DONE

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

DONE

Placement text: Please remove all figure and table placement text. All figures and tables must be cited in the manuscript in numerical order, so please check that they are all still cited in order once placement text is removed.

DONE

Authors' contributions: Please reformat, using the following format as an example, including using initials to refer to each author's contribution and stating that all authors read and approved the final manuscript:

?AB carried out the molecular genetic studies, participated in the sequence
alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

DONE

~

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Key words: Please remove.
DONE

Abbreviations: Please format these in to a paragraph.
DONE

References: the reference list should contain up to 30 authors’ names for each citation. The term “et al.” should not be used if there are fewer than 30 authors.
DONE

Additional file: Please remove, this is not suitable for the additional files section. Please see ‘Covering Letter’ section of the submission system.
DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
DONE (One word was missing)